Author's response to reviews

Title: Ischemic stroke incidence in Santa Coloma de Gramenet (ISISCOG), Spain. A community-based study.

Authors:

M Teresa Alzamora (maiteal2007@gmail.com)
Marta Sorribes (martasorribes.pbcn@ics.scs.es)
Antonio Heras (aheras.bnm.ics@gencat.net)
Marisa Vicheto (mlvicheto.bnm.ics@gencat.net)
Rosa Fores (rfores.bnm.ics@gencat.net)
Jose Sanchez-Oranguren (jsanchez@hes.scs.es)
Nicolas Vila (med020228@nacom.es)
Amparo Sancho (asanchop.bnm.ics@gencat.net)
Guillem Pera (gpera.bnm@ics.scs.es)

Version: 4 Date: 18 March 2008

Author's response to reviews: see over
Dear Sir/Madame;

Please find attached the revised manuscript (1165593437156610) entitled “Ischemic stroke incidence in Santa Coloma de Gramenet (ISISCOG), Spain. A community-based study

We need to change some details of the manuscript:

We have removed the text “BMC Neurology” from the manuscript. We have changed:” The list of authors in the manuscript should be written exactly as they are in the submission system.

We have changed:” Each affiliation in the next format: department, institute, city, and country.

We have included:" a Conclusions header above the paragraph which begins 'In summary...'

We have modified the next reference list 7, 14, 17, 24, 26, 27, 35, 36,

We have changed:”the text black in the reference 15”. We have removed:" the blank pages from the manuscript”..

Thank you very much for your kind attention. I am looking forward to hearing from you.

Yours sincerely,

MT Alzamora Sas
Unitat de Suport a la Recerca
Àmbit Barcelonés Nord i Maresme
CAP El Maresme
C/ Camí del Mig, 36
08303-Mataró
Barcelona (Spain)

Phone: +34 93 741 53 38
Fax: +34 93 799 93 17
E-mail: maiteal2007@gmail.com